Author’s response to reviews

Title: Use of complementary and alternative medicine and self-tests by coronary heart disease patients.

Authors:

  Sheila M Greenfield (s.m.greenfield@bham.ac.uk)
  Helen M Pattison (h.m.pattison@aston.ac.uk)
  Kate B Jolly (c.b.jolly@bham.ac.uk)

Version: 6 Date: 24 July 2008

Author’s response to reviews:

23 July 2008

The BioMed Central Editorial Production Team

Dear Sir

Use of complementary and alternative medicine and self-tests by coronary heart disease patients

I am sending a revised version of the manuscript with the following revisions:-

1. The reference to Table 2 in line 20 of the Methods section has been removed. The first Table mentioned is now Table 1 and the order of Tables mentioned in the text now corresponds to the order of Tables at the end of the manuscript.

2. Text has been removed from the header

3. Tables have been reformatted as requested

Yours sincerely

Dr SM Greenfield

23 July 2008

The BioMed Central Editorial Production Team

Dear Sir

Use of complementary and alternative medicine and self-tests by coronary heart disease patients
Use of complementary and alternative medicine and self-tests by coronary heart disease patients

I am sending a revised version of the manuscript with the following revisions:-
1. The reference to Table 2 in line 20 of the Methods section has been removed. The first Table mentioned is now Table 1 and the order of Tables mentioned in the text now corresponds to the order of Tables at the end of the manuscript.
2. Text has been removed from the header
3. Tables have been reformatted as requested

Yours sincerely

Dr SM Greenfield

23 July 2008

The BioMed Central Editorial Production Team

Dear Sir

Use of complementary and alternative medicine and self-tests by coronary heart disease patients

I am sending a revised version of the manuscript with the following revisions:-
1. The reference to Table 2 in line 20 of the Methods section has been removed. The first Table mentioned is now Table 1 and the order of Tables mentioned in the text now corresponds to the order of Tables at the end of the manuscript.
2. Text has been removed from the header
3. Tables have been reformatted as requested

Yours sincerely

Dr SM Greenfield

23 July 2008

The BioMed Central Editorial Production Team

Dear Sir

Use of complementary and alternative medicine and self-tests by coronary heart disease patients

I am sending a revised version of the manuscript with the following revisions:-
1. The reference to Table 2 in line 20 of the Methods section has been removed.
Dear Sir

Use of complementary and alternative medicine and self-tests by coronary heart disease patients

I am sending a revised version of the manuscript with the following revisions:-

1. The reference to Table 2 in line 20 of the Methods section has been removed. The first Table mentioned is now Table 1 and the order of Tables mentioned in the text now corresponds to the order of Tables at the end of the manuscript.

2. Text has been removed from the header

3. Tables have been reformatted as requested

Yours sincerely

Dr SM Greenfield

23 July 2008

The BioMed Central Editorial Production Team